Author's response to reviews

Title: Effective antiprotease-antibiotic treatment of experimental anthrax

Authors:

Serguei G Popov (spopov@gmu.edu)
Taissia G Popova (tpopova@gmu.edu)
Svetlana Hopkins (svetlana.hopkins@analex.com)
Raymond S Weinstein (rweinstein@gmu.edu)
Rebecca MacAfee (rmacafee@potomach.com)
Karl J Fryxell (kfryxell@gmu.edu)
Vikas Chandhoke (vchandho@gmu.edu)
Charles Bailey (cbailey2@gmu.edu)
Ken Alibek (kalinbek@gmu.edu)

Version: 5 Date: 4 April 2005

Author's response to reviews:

Dear Editor,

Please find uploaded the final version of the manuscript. We made all suggested formatting changes: corrected the title case, cropped figures, added the country on the title page. The reviewer suggested no corrections to the revised manuscript, therefore only a few typos were corrected.